Author’s response to reviews

Title: AIMS65 scoring system is comparable to Glasgow-Blatchford score or Rockall score for prediction of clinical outcomes for non-variceal upper gastrointestinal bleeding

Authors:
Min Seong Kim (kingtta21@naver.com)
Jeongmin Choi (doct00@hanmail.net)
Won Chang Shin (wcshin@paik.ac)

Version: 4 Date: 18 Jul 2019

Author’s response to reviews:

1. Please remove the consent form from the additional files. This form is not required.
Response: We removed the consent form

2. We note that the current submission contains some textual overlap with other previously published works, in particular:

   This overlap mainly exists in the Background and Statistical analysis sections. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.

Response: We did change the format and rephrase the word in the background, statistical analysis, and method section.

3. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.

Response: We submitted only final version of the manuscript.